Author's response to reviews

Title: Comparing early years and childhood experiences and outcomes in Scotland, England and three city-regions: a plausible explanation for Scottish 'excess' mortality?

Authors:

Martin Taulbut (martintaulbut@nhs.net)
David Walsh (David.Walsh@glasgow.gsx.gov.uk)
John O'Dowd (johnodowd@nhs.net)

Version: 5
Date: 30 September 2014

Author's response to reviews:

Dear Editors

Thank you for your guidance in making the final revisions before publication. I have referred the reader to the additional files in the text and added a paragraph briefly describing their content, and uploaded revised manuscript and files accordingly.

I hope that this meets your requirement, but if not, please don’t hesitate to contact me.

Many thanks

Martin